Author’s response to reviews

Title: RE-ALMing Conferences: Evaluating the Adoption, Implementation and Maintenance of the Rick Hansen Institute’s Praxis 2016

Authors:

Heather Gainforth (heather.gainforth@ubc.ca)
Kristy Baxter (kristy.baxter@ubc.ca)
Justine Baron (justinebaron@gmail.com)
Emilie Michalovic (emilie.michalovic@mail.mcgill.ca)
Jeffrey Caron (jeffrey.caron@mail.mcgill.ca)
Shane Sweet (shane.sweet@mcgill.ca)

Version: 1 Date: 05 Dec 2018

Author’s response to reviews:

We thank the editor for the opportunity to revise our manuscript. Specifically, we were asked to:

1. Please rework and resubmit your article paying attention to the various omissions in the manuscript (e.g. all evaluators are referred as 'XX').

Evaluator initials have been added to the manuscript.

2. Providing a greater emphasis on what others can learn from the very specific experience of Praxis.

We have added content about the value of learning about the Praxis initiative in both the introduction and the discussion. Specifically, for each of our main findings outlined in the discussion, we provide a recommendation to inform other large-scale, event-based KT initiatives.